Author’s response to reviews

Title: Prognostic significance of urokinase plasminogen activator and plasminogen activator inhibitor-1 mRNA expression in lymph node- and hormone receptor-positive breast cancer

Authors:

Philippe Leissner (Philippe.leissner@eu.biomerieux.com)
Thibault Verjat (Thibault.verjat@eu.biomerieux.com)
Thomas Bachelot (Bachelot@lyon.fnclcc.fr)
Malick Paye (Malick.paye@eu.biomerieux.com)
Alexander Krause (Alexander.krause@eu.biomerieux.com)
Alain Puisieux (Puisieux@lyon.fnclcc.fr)
Bruno Mougin (Bruno.mougin@eu.biomerieux.com)

Version: 4 Date: 28 August 2006

Author’s response to reviews:

Dear editorial team

As requested, we have modified the title by removing all unnecessary capitals.

With regard to the acknowledgements, all the people who participated in this study or who were involved in drafting or revising the manuscript are referenced as authors. Nobody has to be acknowledged in addition to those authors. So, we suggest not to include the "Acknowledgement section" in the article.

We hope that the manuscript is now acceptable for publication

Best regards,
Philippe Leissner